Author's response to reviews

Title: Frequency, factors and costs associated with injection site infections: findings from a national multi-site survey of injecting drug users in England.

Authors:

Vivian Hope (vivian.hope@hpa.org.uk)
Jo Kimber (Jo.Kimber@lshtm.ac.uk)
Peter Vickerman (peter.vickerman@lshtm.ac.uk)
Matthew Hickman (matthew.hickman@bristol.ac.uk)
Fortune Ncube (fortune.ncube@hpa.org.uk)

Version: 4 Date: 2 September 2008

Author's response to reviews:

As requested the following have been added to the first paragraph of the methods section: details of consent (in sentence 5) and ethical approval (an extra last sentence).